RESEARCH ARTICLE

Evaluation of *Aedes aegypti* control intervention with pyriproxyfen by lcWGS in Manacapuru, Amazonas, Brazil

Lorena Ferreira de Oliveira Leles ^{1,2}*, Marcus Vinicius Niz Alvarez³, Jose Joaquin Carvajal Cortés¹, Diego Peres Alonso³, Paulo Eduardo Martins Ribolla³, Sérgio Luiz Bessa Luz^{1,2}

1 Laboratório de Ecologia de Doenças Transmissíveis na Amazônia, Instituto Leônidas e Maria Deane— Fiocruz Amazônia, Manaus, Brasil, 2 Programa de Pós-Graduação em Biologia Parasitária, Instituto Oswaldo Cruz (IOC), Rio de Janeiro, Brasil, 3 Universidade do Estado de São Paulo (UNESP)—Instituto de Biotecnologia e Biociências, Botucatu, Brasil

* lorenaleless@gmail.com

Abstract

Background

Ae. aegypti mosquitoes are considered a global threat to public health due to its ability to transmit arboviruses such as yellow fever, dengue, Zika and Chikungunya to humans. The lack of effective arboviral vaccines and etiological treatments make vector control strategies fundamental in interrupting the transmission cycle of these pathogens. This study evaluated *Ae. aegypti* mosquito populations pre- and post-intervention period with disseminating stations of the larvicide pyriproxyfen to understand its potential influence on the genetic structure and population diversity of these vectors.

Methodology/Principal findings

This study was conducted in Manacapuru city, Amazonas, Brazil, where 1,000 pyriproxyfen dissemination stations were deployed and monitored from FEB/2014 to FEB/2015 (pre-intervention) and AUG/2015 to JAN/2016 (post-intervention). Low-coverage whole genome sequencing of 36 individuals was performed, revealing significant stratification between pre- and post-intervention groups (pairwise F_{ST} estimate of 0.1126; p-value < 0.033). Tajima's D estimates were -3.25 and -3.07 (both p-value < 0.01) for pre- and post-intervention groups, respectively. Molecular diversity estimates (Theta(S) and Theta(Pi)) also showed divergences between pre- and post-intervention groups. PCA and K-means analysis showed clustering for SNP frequency matrix and SNP genotype matrix, respectively, being both mainly represented by the first principal component. PCA and K-means clustering also showed significant results that corroborate the impact of pyriproxyfen intervention on genetic structure populations of *Ae. aegypti* mosquitoes.

Conclusions/Significance

The results revealed a bottleneck effect and reduced mosquito populations during intervention, followed by reintroduction from adjacent and unaffected populations by this vector. We



OPEN ACCESS

Citation: Leles LFdO, Niz Alvarez MV, Cortés JJC, Alonso DP, Ribolla PEM, Luz SLB (2024) Evaluation of *Aedes aegypti* control intervention with pyriproxyfen by IcWGS in Manacapuru, Amazonas, Brazil. PLoS Negl Trop Dis 18(10): e0012547. https://doi.org/10.1371/journal. pntd.0012547

Editor: Paul O. Mireji, Kenya Agricultural and Livestock Research Organization, KENYA

Received: February 7, 2024

Accepted: September 18, 2024

Published: October 3, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0012547

Copyright: © 2024 Leles et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The authors confirm that all data underlying the findings are fully available without restriction. Data is available at

NCBI with the following BIOPROJECT number PRJNA970519.

Funding: The author(s) received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

highlighted that low-coverage whole genome sequencing can contribute to genetic and structure population data, and also generate important information to aid in genomic and epidemiological surveillance.

Author summary

The Aedes aegypti mosquito is a global public health threat, transmitting diseases like yellow fever, dengue, zika and Chikungunya, requiring several vector control strategies. Traditional methods such as active search and treatment of accessible breeding sites by control agents can be supplemented using PPF dissemination stations. The tactic uses adult mosquitoes themselves to disseminate the larvicide adhered to their bodies, between treated and untreated breeding sites, which are usually cryptic and inaccessible. This tactic can easily be incorporated into integrated vector management strategies, reducing the emergence of adult mosquitoes and, consequently, arboviruses transmission. However, its impact on Ae. aegypti genetic and population structure remain unknown. The PPF intervention caused a bottleneck in the genetic structure, an event in which a population undergoes a drastic reduction in population density, in this case due to the pressure caused by the larvicide, followed by reintroduction of adult mosquitoes from nearby areas not exposed to PPF. This study uses a cost-effective low-coverage whole genome sequencing method to analyze Ae. aegypti genetic population structure based on mitochondrial genome, comparing pre- and post-intervention periods with pyriproxyfen. This approach can reveal both short and long-term effects of this control strategy in the vector genome dynamics. These findings have direct implications for planning control activities, enhancing genomic surveillance and epidemiological surveillance, contributing to global efforts against vector-borne diseases.

Introduction

Currently *Ae. aegypti* is considered a global threat to public health not only due to its aggressiveness (i.e. discomfort caused by the bites) and, invasiveness (i.e. occurrence in wild and urban areas), but also to its ability to transmit arboviruses to humans, such as yellow fever, dengue, Zika and Chikungunya [1,2]. Recently, the Zika virus outbreak and its association with serious damage to the nervous system, including microcephaly, have resulted in national and international public health emergencies [3].

The lack of effective arboviral vaccines (except for yellow fever) and etiological treatments make vector control strategies crucial to interrupt the transmission cycle of these pathogens, which has been shown effective against other vector-borne diseases, such as malaria [4]. In fact, the World Health Organization (WHO) estimates that in the next fifteen years, about 50–60% of investment funds for controlling and eliminating neglected tropical diseases will be allocated to developing new insecticides and transmission control strategies. Understanding the impact of these strategies on vector genomes and exploring cost-effective alternatives is essential, because this approach enables active participation of developing countries, where most of these diseases are prevalent, in the control and elimination efforts [5].

Traditional methods for controlling *Ae. aegypti* include the use of adulticides, active search, and treatment of accessible breeding sites by health control agents [6,7]. Despite the benefits obtained using these methods, one of the main challenges faced in these strategies is the low

efficiency of breeding sites discovery and treatment. Since *Ae. aegypti* oviposition characteristically occurs in small, cryptic, and inaccessible breeding sites, these can go unnoticed during vector control campaigns involving insecticide application [7,8].

An effective strategy to increase the number of treated breeding sites leverages the dissemination of potent larvicides such as pyriproxyfen (PPF), by the mosquitoes themselves, into difficult-to-access breeding sites, interfering in larvae and pupae growth, preventing the vector development [9,10]. The use of this larvicide in disseminating stations (DSs) has proven to effectively reduce the transmission of arboviruses due to the spread of the larvicide by the adult female mosquito during oviposition in untreated breeding sites, consequently resulting in a smaller number of competent vectors available in the environment [4,8]. Despite being effective, the impact of this control strategy on the biological dynamics, genetic and population structure of these vectors remains unknown [8,11].

For *Ae. aegypti*, few studies correlated genetic and population structures with robust molecular techniques and a rigorous epidemiological database and most of these studies identify genetic variations, their distribution and frequency among populations using microsatellite markers and primer-specific SNP genotyping [2,12–14]. Recent advancements in new sequencing techniques, such as *low-coverage whole genome sequence* (lcWGS), have provided extensive data that enable a better understanding of structural dynamics and population genetics of several organisms. Therefore, these approaches allow the evaluation of different control strategies by identifying significant microvariations in populations of important vectors, also allowing genome assembly, gene panels construction and search for SNPs of interest, as already observed for *Anopheles darlingi* [15].

These new sequencing techniques brought significantly reduced costs in population genetics studies, but genomic sequencing expenses remain high for population genetics studies due the large number of samples that need to be sequenced [16]. A viable alternative for large-scale genomic sequencing is the lcWGS, which allows obtaining sufficient genomic data for understanding population genetic structure, associating high reliability and low cost [16,17]. Despite the relatively large genome size of *Ae. aegypti*, lcWGS can still effectively recover mitochondrial genome markers, even when the average depth of nuclear genome sequencing is limited [18,19].

In this paper, we applied a cost-effective, *low-coverage whole genome sequencing* method to evaluate *Ae. aegypti* mosquito populations during the pre- and post-intervention periods with disseminating stations of the larvicide pyriproxyfen [11,16]. This approach can help us understand its potential influence on the genetic structure and population diversity of these vectors, based on long-term exposure to PPF. These findings have direct implications for planning control activities, enhancing genomic and epidemiological surveillance, contributing to global efforts against vector-borne diseases [17,20].

Material and methods

Ethics approval and consent to participate

Dr. Sérgio Luiz Bessa Luz is permanently licensed (27733–1) by Brazilian Institute of Environment and Natural Resources (IBAMA) to collect disease vectors. All procedures were conducted with residents' permission. Formal approval was not required for mosquito collection in urban environments. Ethical review and approval were waived for this study.

Study area and mosquito surveillance

Samples used in this paper are from a previously developed project titled "Insecticide Spread by Mosquitoes for Vector Control and Potential Blocking of Arbovirus Epidemics"⁸. In this study, we conducted a previous intervention study using DSs contaminated with pyriproxyfen in the

Amazonas State. Collections were carried out in the municipality of Manacapuru (Fig 1), Central Amazon, Brazil, a city with approximately 60,000 inhabitants and 13,500 dwellings (data from Brazilian Institute of Geography and Statistics–IBGE).

One hundred dwellings were randomly selected to produce the maximum coverage of the study area. One thousand pyriproxyfen dissemination stations were placed across the city, consisting of black plastic cups and water to attract *Aedes* mosquitoes. DSs were verified twice



Fig 1. Study site: The city of Manacapuru, Amazonas, Brazil. The orange circles indicate the location of the dwellings monitored for mosquito vectors. The map was drawn using the maps library on software QGIS (www.ibge. gov.br/geociencias/organizacao-do-territorio/malhas-territoriais/15774-malhas.html).

https://doi.org/10.1371/journal.pntd.0012547.g001

a month from FEB/2014 to FEB/2015 (pre-intervention period), MAR/2015 to JUL/2015 (intervention) and AUG/2015 to JAN/2016 (post-intervention period) as shown in Fig 2.

The juvenile specimens of *Ae. aegypti* were collected, morphologically identified and later emerging in adult mosquitoes in laboratory conditions. The samples collected were stored in 100% alcohol and placed in a freezer (4°C). Specimens were selected by pre- and post-intervention period, also considering rainy and dry seasonal periods.

Sample preparation and sequencing

For DNA extraction, each specimen was extracted individually using a DNeasy Blood & Tissue (Qiagen, USA) and DNA quantification was performed by fluorometric quantitation using a Thermo Fisher Scientific QuBit dsDNA HS Assay Kit, both according to the manufacturer's recommendations.

DNA libraries were prepared using one-fifth of the total recommended volume for the Nextera XT Library prep kit (Illumina) according to the manufacturer's recommendations. DNA samples were multiplexed, loaded on a mid-output flow cell and sequenced using the Next-Seq500 platform (Illumina) in a 151-cycle single-read run. Sequence-quality analyses were performed using the FASTQC program [21] and reads were used if results from all analysis modules were approved without errors.

Species identification

Sequencing data was aligned with the *Ae. aegypti* cytochrome oxidase subunit I (COI) reference sequence (available at KC913582.1) using Burrows-Wheeler Aligner (BWA) software [22]. After alignment, the individual COI consensus sequences were generated using the Sam-Tools software package [23]. The BLASTn tool was used for multi-species identification using the individual generated COI consensus sequence [24]. Only the highest matching result from BLAST was used. Specimens were discarded if e-value > $1e^{-100}$, identities < 200, and identity < 90%, and if the matching sequence was not identified as *Ae. aegypti*.

Variant calling

Sequencing data quality analysis was performed using FASTQC [21] and reads filtering, trimming and adapter removal procedures were performed using Trimmomatic [25], dropping





https://doi.org/10.1371/journal.pntd.0012547.g002

reads with length less than 80 base pairs. All sequencing reads were aligned against the reference mitochondrial genome of *Ae. aegypti* (NCBI NC_035159.1, available from GenBank) using the BWA program [22]. Variant calling was performed with the SamTools software package [23], and the panel of variants was exported in the VCF version 4.2 format. SNPs were removed from the variant panel based on a minimum allele frequency (MAF) < 0.1 and missing data (MD) > 0.5 using the LCVCFtools program [26]. Genotypes with sequencing depth (DP) < 5 or phred genotype-quality score (GQ) < 20 were filtered.

Mitochondrial population genetics inferences

Statistical analyses were performed with PLINK 1.9 [27] and graphs and figures were generated using the GGPlot2 package in RStudio [28]. Pairwise F_{ST} , nucleotide diversity (θ_S and θ_π) and Tajima's D were estimated using ARLEQUIN 3.5 [29]. The principal component analysis (PCA) procedure was calculated by using both the SNP frequencies estimated for each group and pairwise genetic distance IBS (Identity-by-state matrix). Hierarchical clustering analysis was performed for both PCA results using pvclust v2.2–0 package for R [30].

Results

Data collection and DNA extraction

During the monthly checks from 2014 to 2016, 1.933 specimens of *Ae. aegypti* were collected and morphologically identified, being 455 samples from the dry seasonal period and 1.478 samples from the rainy seasonal period. A total of 36 individuals were submitted to low coverage sequencing, followed by species identification: 18 individuals from the pre-intervention and 16 individuals from the post-intervention period, with half of the samples corresponding to the dry and rainy season from each intervention group. All individuals were identified as *Ae. aegypti*.

Low-coverage whole genome sequencing (lcWGS) performance

A total of 6.705 reads were properly mapped with the mitogenome reference. The sequencing GC content was 21.03%. The sequencing depth (Fig 3) and mapping coverage distribution (Fig 4) are represented below. The average sequencing depth was approximately 1.43 ± 0.26 . After variant calling, the final dataset consisted of 97 SNPs distributed throughout the mitogenome. The SNP average sequencing depth and quality was 4.00 ± 0.71 and 35.9 ± 3.73 , respectively. The SNP density was 5.7 SNP/Kbp.

Statistical and structural analysis

Significant stratification was observed between pre-intervention and post-intervention groups, with a pairwise F_{ST} estimate of 0.1126 (p-value < 0.033, number of permutations = 1000). The estimates of Tajima's D of pre-intervention and post-intervention groups were -3.25 (p-value < 0.01) and -3.07 (p-value < 0.01) respectively. Theta(S) molecular diversity estimates for the pre-intervention and post-intervention groups were 24.7 ± 8.8 and 10.2 ± 3.9, respectively. Theta(Pi) molecular diversity estimates for pre-intervention and post-intervention and post-intervention groups were 5.5 ± 3.1 and 2.9 ± 1.6, respectively.

The PCA analysis showed clustering for SNP frequency matrix, as the figure below (Fig 5). The clustering is mainly represented by the first principal component which corresponds to at least half of the total variance by the model. The SNP rotation estimates were relatively homogeneous. The k-means clustering also showed significant clusters (Fig 6), given the approximately unbiased p-value estimates > 95% (alpha < 0.05).



Fig 3. Mitogenome sequencing coverage boxplot diagram given the respective depth.

The PCA analysis showed clustering for SNP genotype matrix, as the figure below (Fig_7). The clustering is mainly represented by the first principal component which corresponds to the major explained variance by the model.

Discussion

This study provides a comprehensive analysis of the genetic structure of field-collected *Ae*. *aegypti* in Manacapuru, Amazonas, Brazil, before and after intervention involving disseminating stations of pyriproxyfen. From an epidemiological perspective, the intervention led to a significant decrease in juvenile catch, from an average of 3.20 individuals per sentinel/month before the intervention to less than one juvenile during dissemination. Moreover, the initial monthly mortality rate of 6.6% for *Ae. aegypti* increased dramatically to an average of 96% during the intervention period [8]. The combination of reduced juvenile capture due to the near absence of mosquitoes population, and heightened juvenile mortality resulting from exposure to PPF, led to a significant decline in adult mosquito emergence during the intervention period. The effect was particularly drastic, considering the nearly complete absence of mosquitoes due to the intervention [7,8].

Based on Tajima's D estimate, there is indicative evidence of a population expansion, potentially following a recent bottleneck effect triggered by the application of PPF. Stratification analysis (F_{ST}) reveals moderate stratification between pre- and post-intervention groups. Interestingly, both groups presented very similar signs of stratification, considering both the level of stratification and number of permutations. To our knowledge, there are currently no



Fig 4. The reference mitogenome. SNPs are highlighted as black vertical lines. Blue represents gene regions located in the forward strand. Red represents gene regions located in the reverse strand. The inner bar plot represents per-base sequencing depth.

studies that have assessed the genetic structure of field populations of *Ae. aegypti* within the context of a specific intervention.

Following the intervention, there was a slight but significant decrease in nucleotide diversity, as evidenced by the population parameters Theta(S) and Theta(Pi). Moreover, the mitochondrial data for *Ae. aegypti* revealed statistically significant moderate stratification, indicating population differentiation between the pre- and post-intervention groups. Additionally, clustering analysis based on the allele frequency matrix revealed distinct clusters



Fig 5. PCA biplot based on the groupwise SNP frequency matrix. Values between parentheses represent the variance explained by the respective principal component. Grey points represent SNP eigenvectors. Between-groups colored dashed lines represent significant hierarchical clustering groups.

between pre- and post-intervention groups, particularly along the first principal component, which indicates genetic variation associated with the intervention period. These groupings were statistically significant, as confirmed by the approximately unbiased p-value, which indicates significant clustering between the two groups considering their intervention period.

The PCA and hierarchical clustering analysis indicated a substantial genetic distance when comparing individuals from the pre- and post-intervention periods, indicating the presence of two distinct clusters. Checking these clusters, principal component 1 (PC1) demonstrates that there is a greater difference between intervention groups (51.2%) than between seasonal groups on principal component 2 (26.0%), based on the SNP frequency matrix (Fig 5). Furthermore, the identified clusters suggest that individuals from the post-intervention period showed a smaller *within* genetic distance compared to the pre-intervention period (Fig 7), converging with the results from nucleotide diversity estimates. Notably, no clustering was observed for the seasonal period variable, indicating that seasonality had no effect on the clustering analysis.

Based on the analysis described here, the scenario suggests that intervention induced a bottleneck effect on this population. This effect explains that there is an event in which a



Cluster dendrogram with p-values (%)

Distance: correlation Cluster method: average

Fig 6. Hierarchical clustering analysis and multiscale bootstrap resampling based on groupwise SNP frequency matrix. Temporal periods: pre- (before intervention) and post- (after intervention). Seasonal periods: rainy and dry. AU (red): approximately unbiased p-value. BP (green): bootstrap probability.

https://doi.org/10.1371/journal.pntd.0012547.g006

population undergoes a drastic reduction in population density and subsequent population growth, in this case due to the pressure caused by the larvicide PPF and reintroduction of adult *Ae. aegypti* mosquitoes from nearby areas not exposed to PPF [31,32]. This is corroborated by the drastic reduction in the number of individuals in this region following the PPF intervention, wherein individuals from the pre-intervention period were nearly eliminated during larvicide application. Subsequently, the reintroduction from adjacent populations unaffected by PPF intervention occurred. Moreover, these data demonstrated that lcWGS is a highly effective and promising technique, not only due to its cost-effectiveness but also its capability to recover mitochondrial genome information at very low coverage.

The development of new sequencing approaches, such as whole genome sequencing (WGS) and double digest restriction-site associated DNA sequencing (ddRADseq), offers high sequencing depth and coverage of sequenced samples, but also comes with high execution costs. The lcWGS is a sequencing alternative for studies of population genetic structure, providing reliable data for marker selection, and is more economical than other techniques [17,33]. Additionally, there is a considerable cost reduction since our protocol uses one-fifth of the total volume recommended by the manufacturer for library preparation, and reducing the





coverage from 20X to 1X, while simultaneously using a larger number of samples and data processing to increase the statistical power of our analyses [33,34]. This enables us to investigate the genetic structure and population dynamics of *Ae. aegypti*. lcWGS has proven to be a valuable technique for genetically characterizing populations of other medically important vectors belonging to the Culicinae subfamily [35], such as *Nyssorhynchus darlingi*, at a microgeographical scale and susceptibility to *Plasmodium* spp. infection [15,19,36].

Our findings also contribute significantly to emerging laboratory approaches such as genomic surveillance, within epidemiological surveillance, as a valuable tool using information from the genomes of any pathogen and/or its vectors for disease monitoring and control across diverse scenarios [37]. It aids in monitoring insecticide resistance, by revealing genetic mutations associated with it and avoiding the use of insecticides to which vectors have developed resistance. Also aids to development of personalized strategies, by identifying genetic diversity, which becomes possible to develop specific control strategies considering characteristics of local vector populations and increasing the effectiveness of intervention approaches [37,38]. Furthermore, it can be a useful tool for tracking arboviruses dissemination routes (i.e. tracking *Ae. aegypti* dispersion to understand the transmission of diseases such as dengue, Zika and Chikungunya) and predicting and preventing epidemics and potential outbreaks (i.e. allowing health authorities to seek an effective response, before arboviruses cases reach alarming proportions) [39–41].

Despite the challenges and need for further studies, large-scale genomic surveillance emerges as a promising tool, considering the use of high-throughput genomic sequencing and epidemiological conditions to understand not only the previously mentioned issues, but also the genetic structure and population dynamics of several vectors of medical importance.

Conclusion

Population genetic structure observed in this scenario, comparing *Ae. aegypti* specimens before and after the PPF intervention suggest a bottleneck effect, leading to a reduction in the mosquito population in this region. Subsequently, once the intervention was removed, the region experienced reintroduction by adjacent populations of *Ae. aegypti* unaffected by PPF exposure. These findings suggested that lcWGS consists in a valuable tool for genetically characterizing diverse vector populations in association with intervention periods and approaches, considering not only the cost-effectiveness of this technique, but also its contribution and importance to genomic surveillance. Notably, to the best of our knowledge, this study is the first to use field-collected *Ae. aegypti* to evaluate the genetic structure of individuals exposed to intervention methods recommended by the national dengue control program (PNCD). Despite the limitations, such as the low coverage of nuclear markers, additional studies using lcWGS are needed to better characterize the genetic structure of the vector mosquito population associated with genomic surveillance, especially for *Aedes aegypti*.

Acknowledgments

We are grateful to Leônidas and Maria Deane Institute (Fiocruz Amazônia), Oswaldo Cruz Institute (Fiocruz) and Sao Paulo State University "Júlio de Mesquita Filho" (UNESP-Botucatu), who contributed to this work.

Author Contributions

Conceptualization: Lorena Ferreira de Oliveira Leles, Sérgio Luiz Bessa Luz.

Data curation: Jose Joaquin Carvajal Cortés, Sérgio Luiz Bessa Luz.

Formal analysis: Lorena Ferreira de Oliveira Leles, Marcus Vinicius Niz Alvarez.

Funding acquisition: Sérgio Luiz Bessa Luz.

Investigation: Lorena Ferreira de Oliveira Leles, Jose Joaquin Carvajal Cortés.

Methodology: Lorena Ferreira de Oliveira Leles.

Project administration: Paulo Eduardo Martins Ribolla, Sérgio Luiz Bessa Luz.

Resources: Paulo Eduardo Martins Ribolla, Sérgio Luiz Bessa Luz.

Software: Marcus Vinicius Niz Alvarez.

Supervision: Diego Peres Alonso, Paulo Eduardo Martins Ribolla.

Validation: Lorena Ferreira de Oliveira Leles, Marcus Vinicius Niz Alvarez.

Visualization: Lorena Ferreira de Oliveira Leles, Marcus Vinicius Niz Alvarez, Jose Joaquin Carvajal Cortés, Diego Peres Alonso.

Writing - original draft: Lorena Ferreira de Oliveira Leles.

Writing – review & editing: Lorena Ferreira de Oliveira Leles, Marcus Vinicius Niz Alvarez, Diego Peres Alonso, Paulo Eduardo Martins Ribolla, Sérgio Luiz Bessa Luz.

References

- LAMBRECHTS L. and SCOTT T.W. Mode of transmission and the evolution of arbovirus virulence in mosquito vectors. Proc Biol Sci., 276–1660:1369–1378, 2009. https://doi.org/10.1098/rspb.2008.1709 PMID: 19141420
- EVANS BR.; GLORIA-SORIA A.; HOU L.; McBRIDE C.; BONIZZONI M.; ZHAO H.; POWELL JR. A multipurpose, high-throughput single-nucleotide polymorphism chip for the dengue and yellow fever mosquito, *Aedes aegypti.* G3, 26-5(5):711–8, 2013.
- ALBUQUERQUE MFPM.; SOUZA WV.; ARAUJO TVB.; BRAGA MC.; MIRANDA FILHO DB.; XIMENES RAA.; MELO FILHO DA. et al. The microcephaly epidemic and Zika virus: Building knowledge in epidemiology. Cad. Saúde Pública, Rio de Janeiro, v. 34, n. 10:(11–34), 2018.
- BRASIL. Relatório da Reunião Internacional para Implementação de Alternativas para o Controle do Aedes aegypti no Brasil. Boletim Epidemiológico. Secretaria de Vigilância em Saúde – Ministério da Saúde, v. 47, n. 15:(2358–9450), 2016.
- ISGBLOBAL—Annual Report ISGlobal. Barcelona Institute for Global Health. Barcelona (available at https://www.isglobal.org), 2018.
- WEAVER SC. and REISEN WK. Present and future arboviral threats. Antiviral Res. Feb; 85(2):328–45, 2010. https://doi.org/10.1016/j.antiviral.2009.10.008 PMID: 19857523
- ABAD-FRANCH F.; ZAMORA-PEREA E.; FERRAZ G.; PADILLA-TORRES SD.; LUZ SLB. Mosquitodisseminated pyriproxyfen yields high breeding-site coverage and boosts juvenile mosquito mortality at the neighborhood scale. PLoS Negl Trop Dis. Apr 7; 9(4):e0003702, 2015. <u>https://doi.org/10.1371/</u> journal.pntd.0003702 PMID: 25849040
- ABAD-FRANCH F.; ZAMORA-PEREA E.; LUZ SLB. Mosquito-disseminated inseticide for citywide vector control and its potential to block arbovirus epidemics: entomological observations and modeling results from Amazonian Brazil, PLoS Med, 17; 14(1):e1002213, 2017.
- 9. ITOH T.; KAWADA H.; ABE A.; ESHITA Y.; RONGSRIYAM Y.; IGARASHI A. Utilization of bloodfed females of *Aedes aegypti* as a vehicle for the transfer of the insect growth regulator pyriproxyfen to larval habitats. J Am Mosq Control Assoc. Sep- 10(3):344–7, 1994.
- DEVINE GJ.; PEREA EZ.; KILLEEN GF.; STANCIL JD.; CLARK SJ.; MORRISON AC. Using adult mosquitoes to transfer insecticides to *Aedes aegypti* larval habitats. Proc Natl Acad Sci U S A. Jul-14; 106(28):11530–4, 2009.
- SEIXAS G.; SALGUEIRO P.; SILVA AC.; CAMPOS M.; SPENASSATOO C.; REYE-LUGO M.; et al. Aedes aegypti on Madeira Island (Portugal): genetic variation of a recently introduced dengue vector. Mem Inst Oswaldo Cruz, Rio de Janeiro, v. 108, suppl. I, p. 3–10, 2013.
- STEFFLER LM.; DOLABELLA SS.; RIBOLLA PE.; DREYER CS.; ARAÚJO ED.; OLIVEIRA RG.; et al. Genetic variability and spatial distribution in small geographic scale of *Aedes aegypti* (Diptera: Culicidae) under different climatic conditions in Northeastern Brazil. Parasit Vec. Oct- 4; 9(1):530, 2016.
- CAMPOS M.; SPENASSATTO C.; MACORIS MLG.; PADUAN KS.; PINTO P.; RIBOLLA PEM. Seasonal population dynamics and the genetic structure of the mosquito vector *Aedes aegypti* in São Paulo, Brazil (2012). Ecology and Evolution. 2(11): 2794–2802, 2012.
- 14. BRITO LP.; CARRARA L.; FREITAS RM.; LIMA JBP.; MARTINS AJ. Levels of Resistance to Pyrethroid among Distinct *kdr* Alleles in *Aedes aegypti* Laboratory Lines and Frequency of *kdr* Alleles in 27 Natural Populations from Rio de Janeiro, Brazil. Biomed Res Int. Jul-11:2410819, 2018.
- CAMPOS M.; CONN JE.; ALONSO DP.; VINETZ JM.; EMERSON KJ.; RIBOLLA PEM. Microgeographical structure in the major Neotropical malaria vector *Anopheles darlingi* using microsatellites and SNP markers. Parasit Vectors. Feb-13; 10(1):76., 2017.
- GOODWIN S.; MCPHERSON JD.; MCCOMBIE WR. Coming of age: ten years of next-generation sequencing technologies. Nature Reviews Genetics, v. 17, n. 6, p. 333, 2016. <u>https://doi.org/10.1038/ nrg.2016.49</u> PMID: 27184599
- GORJANC G.; DUMASY JF.; GONEN S.; GAYNOR RC.; ANTOLIN R.; HICKEY JM. Potential of Low-Coverage Genotyping-by-Sequencing and Imputation for Cost-Effective Genomic Selection in Biparental Segregating Populations. Crop Science, v. 57, n. 3, p. 1404–1420, 2017.
- MATTHEWS BJ.; DUDCHENKO O.; KINGAN SB.; KOREN S.; ANTOSHECHKIN I.; CRAWFORD JE. et al. Improved reference genome of *Aedes aegypti* informs arbovirus vector control. Nature, 563 (7732):501–507, 2018.

- ALVAREZ MVN.; ALONSO DP.; KADRI SM.; RUFALCO-MOUTINHO P.; BERNARDES IAF.; MELLO ACF. et al. Nyssorhynchus darlingi genome-wide studies related to microgeographic dispersion and blood-seeking behavior. Parasites & Vectors, 15:106, 2022.
- PADILLA-TORRES SD.; FERRAZ G.; LUZ SLB.; ZAMORA-PEREA E.; ABAD-FRANCH F. Modeling dengue vector dynamics under imperfect detection: three years of site-occupancy by *Aedes aegypti* and *Aedes albopictus* in urban Amazonia. PLoS One. 8(3):e58420, 2013.
- 21. ANDREWS, S. FastQC: A Quality Control Tool for High Throughput Sequence Data. (Available at: http://www.bioinformatics.babraham.ac.uk/projects/fastqc/). 2010.
- LI H. and DURBIN R. Fast and accurate short read alignment with Burrows-Wheeler transform. Bioinformatics. Jul-15; 25(14):1754–60. 2009. https://doi.org/10.1093/bioinformatics/btp324 PMID: 19451168
- 23. LI H. A statistical framework for SNP calling, mutation discovery, association mapping and population genetical parameter estimation from sequencing data. Bioinformatics. Nov-1; 27(21):2987–93. 2011. https://doi.org/10.1093/bioinformatics/btr509 PMID: 21903627
- ALTSCHUL SF.; GISH W.; MILLER W.; MYERS EW.; LIPMAN DJ. Basic local alignment search tool. J Mol Biol. Oct-5; 215(3):403–10. 1990. https://doi.org/10.1016/S0022-2836(05)80360-2 PMID: 2231712
- BOLGER AM.; LOHSE M.; USADEL B. Trimmomatic: a flexible trimmer for Illumina sequence data. Bioinformatics. Aug-1; 30(15):2114–20. 2014. https://doi.org/10.1093/bioinformatics/btu170 PMID: 24695404
- 26. ALVAREZ MVN. LCVCFtools v1.0.2-alpha (v1.0.2-alpha), Zenodo, 2021.
- PURCELL S.; NEALE B.; TODD-BROWN K.; THOMAS L.; FERREIRA MA.; BENDER D. et al. PLINK: a tool set for whole-genome association and population-based linkage analyses. Am J Hum Genet. Sep- 81(3):559–75. 2007. https://doi.org/10.1086/519795 PMID: 17701901
- 28. RSTUDIO TEAM. RStudio, Inc., Boston, MA., v. 42, p. 14, 2015.
- 29. EXCOFFIER L. and LISCHER HE. Arlequin suite ver 3.5: a new series of programs to perform population genetics analyses under Linux and Windows. Mol Ecol Resour. May; 10(3):564–7. 2010. <u>https://</u> doi.org/10.1111/j.1755-0998.2010.02847.x PMID: 21565059
- **30.** TEAM, R. Core. RA Lang Environ Stat Comput, v. 55, p. 275–286, 2013.
- 31. GILLESPIE JH. Population genetics: a concise guide. JHU Press, 2004.
- BUTLIN RK. Population genomics and speciation. Genetica 138, 409–418, 2010. <u>https://doi.org/10.1007/s10709-008-9321-3 PMID: 18777099</u>
- SIMS D.; SUDBERY I.; ILOTT NE.; HEGER, A.; PONTING, CP. Sequencing depth and coverage: key considerations in genomic analyses. Nature Reviews Genetics, v. 15, n. 2, p. 121, 2014.
- **34.** WETTERSTRAND KA. DNA sequencing costs: data from the NHGRI Genome Sequencing Program (GSP). National Human Genome Research Institute, 2013.
- SILVA A.F.; MACHADO L.C.; PAULA M.B.; VIEIRA C.J.S.P.; BRONZONI R.V.M.; SANTOS M.A.V.M.; WALLAU G.L. Culicidae evolutionary history focusing on the Culicinae subfamily based on mitochondrial phylogenomics. Scientific Reports, 10:18823, 2020. https://doi.org/10.1038/s41598-020-74883-3 PMID: 33139764
- ALONSO D.P.; ALVAREZ M.V.N.; RIBOLLA P.E.M.; CONN J.E.; OLIVEIRA T.M.P.; SALLUM M.A.M. Susceptibility of Field-Collected Nyssorhynchus darlingi to Plasmodium spp. in Western Amazonian Brazil. Genes, 12, 1693, 2021.
- WORLD HEALTH ORGANIZATION. Global genomic surveillance strategy for pathogens with pandemic and epidemic potential 2022–2032: progress report on the first year of implementation. Geneva, 2023.
- BRASIL. Ministério da Saúde. Política Nacional de Vigilância em Saúde. Resolução no 588/2018 do Conselho Nacional de Saúde (CNS). Política Nacional de Vigilância em Saúde (PNVS). Ministério da Saúde: Brasília/DF, 2018.
- TRAN VT.; INWARD RPD.; GUTIERREZ B.; NGUYEN NM.; NGUYEN PT.; RAJENDIRAN I. et al. Reemergence of Cosmopolitan Genotype Dengue Virus Serotype 2, Southern Vietnam. Emerging Infectious Diseases. Vol. 29, No. 10, October, 2023. <u>https://doi.org/10.3201/eid2910.230529</u> PMID: 37735803
- 40. FRITSCH H.; GIOVANETTI M.; XAVIER J.; ADELINO TER.; FONSECA V.; JESUS JG. et al. Retrospective Genomic Surveillance of Chikungunya Transmission in Minas Gerais State, Southeast Brazil. American Society for Microbiology. Microbiology Spectrum. Vol. 10, Issue 5, October, 2022.
- SANTIAGO GA.; KALINICH CC.; CRUZ-LÓPEZ F.; GONZÁLEZ GL.; FLORES B.; HENTOFF A. et al. Tracing the Origin, Spread, and Molecular Evolution of Zika Virus in Puerto Rico, 2016–2017. Emerging Infectious Diseases. Vol. 27, No. 11, November, 2021. <u>https://doi.org/10.3201/eid2711.211575</u> PMID: 34670646